Author’s response to reviews

Title: Clinical manifestations and anti-TNF alpha therapy of juvenile Behçet’s disease in Taiwan

Authors:

Ya-Chiao Hu (ychu0407@gmail.com)
Yao-Hsu Yang (yan0126@ms15.hinet.net)
Yu-Tsan Lin (yutsanlin@ntu.edu.tw)
Li-Chieh Wang (lcwang5@gmail.com)
Hsin-Hui Yu (hhyu121@gmail.com)
Jyh-Hong Lee (leonid.ljh@gmail.com)
Bor-Luen Chiang (gicmbor@ntu.edu.tw)

Version: 4 Date: 01 Jul 2019

Author’s response to reviews:

Dear Ms. Tillie Cryer and Mr. Brock Madsen:

Enclosed please find the revised manuscript entitled “Clinical manifestations and anti-TNF alpha therapy of juvenile Behçet’s disease in Taiwan”.

We are grateful for the constructive suggestions from the editors. We have done some modification of the manuscript as the forth revised version. The revisions are made in the Declarations part and not highlighted in this version.

1. On page 27, (in Declarations part), we revise the statement of “Availability of data and material” to “The datasets generated and analyzed during the current study are derived from chart review and are not publicly available due to data privacy concerns, but are available from the corresponding author on reasonable request.”

2. On page 28, (in Declarations part), we add the sentence “All authors have read and approved the manuscript” in the end of the “Authors’ contributions” section.
We hope that we have made all the modifications satisfactorily and that this manuscript is now acceptable for publication.

Sincerely

Yao-Hsu Yang, MD. PhD.
Department of Pediatrics, National Taiwan University Hospital,
No. 7 Chung-Shan South Road, Taipei, Taiwan.
TEL: 886-2-2312-3456 Ext. 71521
FAX: 886-2-23147450
E-mail: yan0126@ms15.hinet.net